Author's response to reviews

Title: Effects of mothers' socio-economic status on the management of febrile conditions in their under five children in a resource limited setting.

Authors:

Adenike E Olaogun (nikeolaogun@yahoo.com)
Abayomi A Adebayo (bayoade@oauife.edu.ng)
Olufemi E Ayandiran (olufeman@yahoo.com)
Olayinka A Olasode (olayinkaolasode@yahoo.com)

Version: 15 Date: 20 January 2006

Author’s response to reviews:

The formatting changes have been effected.
1. Title- Changed into sentence case and the full stop at end of title has been removed both in the submission system and in the manuscript.
2. Title and Abstract are identical both in the manuscript and submission system.
3. In the main text, 'Introduction' has been renamed 'Background'. 'Methodology' has also been renamed 'Methods'.
4. The manuscript has been checked for errors.

Thank you.